\nocite{Rocca.sep.32.1}
\nocite{Claerbout.sep.32.37}
\nocite{Toldi.sep.35.03}
\nocite{Toldi.sep.38.39}
\nocite{Toldi.sep.38.89}
\nocite{Toldi.sep.41.77}
\nocite{Toldi.sep.41.99}
\nocite{Toldi.sep.43.0}
\nocite{Toldi.sep.43.9}
\nocite{Toldi.sep.43.33}
\nocite{Toldi.sep.43.69}
\nocite{Toldi.sepphd.43}
\nocite{Toldi.sep.43}
\nocite{Fowler.sep.48.65}


\title{Report bibliography}
This is my bibliography of articles
in the progress reports of the Stanford Exploration Project
written from 1974 to the present.
There are  13 
articles.
{\footnotesize
\putbib[SEP]
}
